# Training teachers in classroom management to improve mental health in primary school children: the STARS cluster RCT

Tamsin Ford, 1\* Rachel Hayes, 1 Sarah Byford, 2 Vanessa Edwards, 1 Malcolm Fletcher, 1 Stuart Logan, 1 Brahm Norwich, 3 Will Pritchard, 4 Kate Allen, 1 Matthew Allwood, 1 Poushali Ganguli, 2 Katie Grimes, 5 Lorraine Hansford, 1 Bryony Longdon, 1 Shelley Norman, 6 Anna Price, 1 Abigail Emma Russell 1 and Obioha C Ukoumunne 7

**Declared competing interests of authors:** Stuart Logan was a member of the Health Technology Assessment (HTA) Commissioning Board 2003–10, the HTA Medicines for Children Themed Call 2005–6 and the Rapid Trials and Add on Studies Board 2012.

Published March 2019 DOI: 10.3310/phr07060

<sup>&</sup>lt;sup>1</sup>University of Exeter Medical School, Exeter, UK

<sup>&</sup>lt;sup>2</sup>King's Health Economics, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK

<sup>&</sup>lt;sup>3</sup>Graduate School of Education, University of Exeter, Exeter, UK

<sup>&</sup>lt;sup>4</sup>Education and Early Years, Cornwall County Council, Truro, UK

<sup>&</sup>lt;sup>5</sup>Educational and Counselling Psychology and Special Education, University of British Columbia, Vancouver, BC, Canada

<sup>&</sup>lt;sup>6</sup>School of Psychology, University of Exeter, Exeter, UK

<sup>&</sup>lt;sup>7</sup>National Institute for Health Research Collaborations for Leadership in Applied Health Research and Care South West Peninsula (PenCLAHRC), University of Exeter, Exeter, UK

<sup>\*</sup>Corresponding author T.J.Ford@exeter.ac.uk

# **Plain English summary**

# The STARS cluster RCT

Public Health Research 2019; Vol. 7: No. 6 DOI: 10.3310/phr07060

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

B etween 5% and 6% of primary school-aged children, an average of one or two in every class, can be described as having extremely challenging behaviour, which can be difficult for teachers to manage and makes it hard for all children in the class to learn effectively. The Supporting Teachers And childRen in Schools (STARS) trial wanted to know if attending Teacher Classroom Management (TCM) training helped teachers to improve their behaviour-management strategies and if this in turn would lead to better mental health for the children they teach. The trial involved 80 primary school teachers, and their classes, across the south-west of England. Half of the teachers attended TCM training (intervention) and half did not (control). The mental health and behaviour of the children they taught was compared before training started and after it had finished, 9 months later. The children continued to be compared for the next two academic years, at 18 and 30 months from the start of the study. A total of 2075 children aged between 4 and 9 years took part, and teachers reported that they enjoyed the training and found it helpful. Children whose teachers attended the training had slightly better mental health at 9 months than children whose teachers had not attended the training. These differences were not found 18 and 30 months later when the children were taught by different teachers; both sets of children scored similarly. Although the difference found was not large, the children who initially had the worst mental health made bigger improvements in their mental health, and these improvements were still present 30 months later. The children in the intervention arm had better classroom-based behaviour at the 9-, 18- and 30-month comparisons. It was also found that the intervention may be better value for money than teaching as usual, particularly at the 9- and 18-month follow-up points. TCM training should be tested further for primary school teachers but the trial indicated that it would work best if it were to involve all teachers and teaching assistants in a school.

## **Public Health Research**

ISSN 2050-4381 (Print)

ISSN 2050-439X (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PHR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/phr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Public Health Research journal

Reports are published in *Public Health Research* (PHR) if (1) they have resulted from work for the PHR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Public Health Research* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### PHR programme

The Public Health Research (PHR) programme, part of the National Institute for Health Research (NIHR), evaluates public health interventions, providing new knowledge on the benefits, costs, acceptability and wider impacts of non-NHS interventions intended to improve the health of the public and reduce inequalities in health. The scope of the programme is multi-disciplinary and broad, covering a range of interventions that improve public health. The Public Health Research programme also complements the NIHR Health Technology Assessment programme which has a growing portfolio evaluating NHS public health interventions.

For more information about the PHR programme please visit the website: http://www.nets.nihr.ac.uk/programmes/phr

#### This report

The research reported in this issue of the journal was funded by the PHR programme as project number 10/3006/07. The contractual start date was in June 2012. The final report began editorial review in March 2018 and was accepted for publication in September 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PHR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Ford et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

## **NIHR Journals Library Editor-in-Chief**

**Professor Ken Stein** Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

### **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

**Professor Martin Underwood** Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk